

Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2022GL098856

Key Points:

- The latest generation of coupled climate models project an increase in severity and frequency of very wet Indian summer monsoon seasons
- Very wet monsoon seasons are projected to occur 8 times more often in 2050–2100 compared to 1965–2015 under unabated climate change
- On the subseasonal scale, there is a shift from days with light rainfall to days with moderate or heavy rainfall

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

A. Levermann, anders.levermann@pik-potsdam.de

Citation:

Katzenberger, A., Levermann, A., Schewe, J., & Pongratz, J. (2022). Intensification of very wet monsoon seasons in India under global warming. *Geophysical Research Letters*, 49, e2022GL098856. https://doi. org/10.1029/2022GL098856

Received 31 MAR 2022 Accepted 19 JUL 2022

© 2022. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Intensification of Very Wet Monsoon Seasons in India Under Global Warming

Anja Katzenberger^{1,2}, Anders Levermann^{1,3,4}, Jacob Schewe¹, and Julia Pongratz^{2,5}

¹Potsdam Institute for Climate Impact Research, Potsdam, Germany, ²Ludwig-Maximilian University, Munich, Germany, ³LDEO, Columbia University, Palisades, NY, USA, ⁴Potsdam University, Potsdam, Germany, ⁵Max Planck Institute for Meteorology, Hamburg, Germany

Abstract Rainfall-intense summer monsoon seasons on the Indian subcontinent that are exceeding long-term averages cause widespread floods and landslides. Here we show that the latest generation of coupled climate models robustly project an intensification of very rainfall-intense seasons (June–September). Under the shared socioeconomic pathway SSP5-8.5, very wet monsoon seasons as observed in only 5 years in the period 1965–2015 are projected to occur 8 times more often in 2050–2100 in the multi-model average. Under SSP2-4.5, these seasons become only a factor of 6 times more frequent, showing that even modest efforts to mitigate climate change can have a strong impact on the frequency of very strong rainfall seasons. Besides, we find that the increasing risk of extreme seasonal rainfall is accompanied by a shift from days with light rainfall to days with moderate or heavy rainfall. Additionally, the number of wet days is projected to increase.

Plain Language Summary The South Asian monsoon affects the life of more than one billion people. In the past, summer monsoon seasons (June–September) with very intense rainfall have been associated with widespread floods and an increased number of landslides. Here, we set the focus on the question how the probability of these very wet monsoon seasons will change in the 21st century under climate change. For this purpose, we use the latest generation of climate models with improved performance regarding the Indian monsoon as well as reduced uncertainties compared to the previous model generation. Under the strongest emission scenario, very wet monsoon seasons that used to be observed in 5 out of 50 years in the period 1965–2015 are projected to occur 8 times more frequently in 2050–2100 on multi-model average. With modest mitigation efforts, this is reduced to a factor of 6 in the future period. Besides, this increase in frequency and intensity of extreme monsoon seasons is accompanied by a shift from days with light rainfall to days with moderate or heavy rainfall. Additionally, the number of wet days is projected to increase. The particular character of the change depends on the determination of humankind to reduce carbon emissions and implement mitigation measures.

1. Introduction

There have been numerous floods in recent years associated with the Indian summer monsoon (ISM) as a component of the South Asian monsoon, for example, the Mumbai floods in 2005 (Bohra et al., 2006), floods in Northwest India and Pakistan in 2010 as a response to a strong La Niña event (Mujumdar et al., 2012) or those in Kerala in 2018 (Vishnu et al., 2019). Since the 1980s, there have been more than 95,000 deaths associated with floods and landslides in the countries of the Indian subcontinent (Guha-Sapir et al., 2018). Hunt and Menon (2020) found that the Kerala flooding event in 2018 could be 36% more rainfall-intense under a RCP8.5 climate. Almazroui et al. (2021) projected an increase in the annual maximum consecutive 5-day precipitation in the Asian monsoon regions and derived a higher risk for extreme flooding in the Asian monsoon regions. Given the agricultural yield's sensitivity to the monsoon rainfall variability and associated extremes (DeFries et al., 2016; Prasanna, 2014; Revadekar & Preethi, 2012), understanding how the ISM responds to global warming is also crucial for crop yields and food security in the region as well as for numerous other aspects of public and individual life, like water management or the country's economy.

The data from various phases of Coupled Model Intercomparison projects (CMIPs) has been widely used to examine the projected changes in the global climate as well as its different components such as the ISM. In the last generation of climate models - which were the basis of earlier studies on seasonal extremes – studies identified monsoon rainfall features that were not yet well represented in the CMIP5 models (Tayler et al., 2012).

Recently, the data from the latest phase 6 of CMIP (CMIP6) (Eyring et al., 2016) has become available and it was confirmed that they bring relevant improvements regarding the ISM's characteristics (Dong & Dong, 2021; Gusain et al., 2020; Rajendran et al., 2021). By updating deep convective schemes, by modifying indirect effects of aerosols in cloud formation as well as by implementing finer resolutions the models have undergone further improvements (Gusain et al., 2020; Rajendran et al., 2021). The improved model capacity in capturing the meridional moisture flux convergence might have contributed to the reduction in dry and wet biases in the Asian monsoon region as well as to the models capacity in reproducing extreme precipitation (Dong & Dong, 2021). Nevertheless, other dynamics as the relationship between the ISM rainfall and the Equatorial Indian Ocean Oscillation are not yet fully captured adequately in many models. Given this relationship's crucial impact on the interannual variability of the ISM, also in this generation of CMIP, there still remains potential for further improvements (Rajendran et al., 2021).

There is a widespread agreement among global climate models that the rainfall during the ISM will increase throughout the 21st century (Almazroui et al., 2020; Chaturvedi et al., 2012; Katzenberger et al., 2021; Lee & Wang, 2014; Mei et al., 2015; Menon et al., 2013; Sharmila et al., 2015). Using 32 models of the latest climate model generation (CMIP6), Katzenberger et al. (2021) quantified the projected summer monsoon rainfall increase to be between 9.7% and 24.3% by the end of the 21st century depending on the underlying emission scenario. Also a linear dependence of the ISM rainfall and the global temperature independent of the scenarios was found and estimated to be 0.33 mm/day per degree Celsius which corresponds to 5.4% of the current annual rainfall (Ha et al., 2020; Katzenberger et al., 2021). Thus, the rainfall of the ISM domain is more sensitive to rising global temperatures than global precipitation (1%–3% per degree Celsius, Ha et al. (2020)). In addition, an increase of the interannual variability of seasonal rainfall is projected (Jayasankar et al., 2015; Katzenberger et al., 2021; Kitoh, 2017; Menon et al., 2013) raising the amount of rainfall in some seasons even further.

While the literature on the projections of the ISM's interannual variability has been converging toward an intensifying tendency, the particular outcome for wet seasonal extremes had only drawn limited attention until recent years, which was the motivation for Kamizawa and Takahashi (2018) to use 22 CMIP5 models to address this research question. In their study, the changes in the wettest (driest) and second wettest (driest) season between 2007–2031 and 2076–2100 were examined under the Representative Concentration Pathway 4.5 (RCP4.5). Most of the CMIP5 models projected that the wet seasonal extremes expand over the Indian subcontinent, but it has to be noted, that focusing on the maxima is a method highly susceptible to the bias of outliers. These results coincide with the earlier study by Sharmila et al. (2015), who found that the years with strong monsoon rainfall are expected to increase in frequency as well as severity by the end of the 21st century but the results still had a very strong inter-model spread. Against this background and given the improvements in CMIP6 compared to CMIP5 as explained before, the results of CMIP6 regarding the seasonal extremes are of particular interest. Additionally, Dong and Dong (2021) point out the improvements between CMIP5 and CMIP6 regarding the daily rainfall amount (exceeding 10 and 20 mm) which was additional motivation for this study to also analyze the question how the changes in seasonal rainfall translate to selected indices on the subseasonal scale.

Here, we use the data of 32 CMIP6 models in order to quantify the changes in seasonal rainfall extremes in India under different emission scenarios. We also analyze how these changes translate to the subseasonal scale, particularly changes of daily rainfall. In order to do so, we select six models with a better monsoon performance in the historical simulations in order to get an insight about their projections following the detailed evaluation provided by Rajendran et al. (2021). Section 2 gives an overview about the underlying scenarios, the CMIP6 data, the definition of seasonal extremes and the characterization of daily rainfall as well as the selection criteria for the models. Section 3 gives a detailed insight about our results and in Section 4 these results are discussed in the context of similar studies.

2. Methods

2.1. CMIP6 Data

For our analysis we include all models that were available for the historic period (1850–2015) as well as the future period (2015–2100) under SSP5-8.5 in ScenarioMIP (O'Neill et al., 2016) which is the combination of the socio-economic scenario pathway 5 (SSP5) and the RCP8.5 (Van Vuuren et al., 2014). We also conduct the analysis for the SSP2-4.5 (SSP2, RCP4.5) and SSP1-2.6 (SSP1, RCP2.6) scenario. Even if for some models several

realizations were available, we only used one ensemble member. We select the area of India and neighboring areas (longitude 67°–98°E and latitude 6°–36°N; see Figure S1 in Supporting Information S1) as ISM region and focus on the land area. The analysis of the wet seasons is performed using the individual grids of each model (For native model resolution, see Table S3 in Supporting Information S1), and for further spatial analysis, the grids of the models with the better ISM performance in the historical simulations are standardized to a $1^{\circ} \times 1^{\circ}$ latitude/longitude grid by first order conservative remapping. In this context, it has to be noted that the native spatial resolution strongly differs between the models ranging from 100 to 500 km over land. For the analysis of the ISM seasons we use monthly data, for the research questions on subseasonal we analyze data on a daily scale. For each model and each summer season, we calculate the mean monsoon rainfall by averaging the monthly rainfall data from June-September (JJAS) over the region of interest to obtain the mean daily precipitation (units: mm/day; please note that e.g., a total rainfall amount of 880 mm per season is equivalent to 880 mm/122 days in season = 7.2 mm/day). In order to compare the rainfall in 2050–2100 to a baseline period, we decided to use 1965-2015 in accordance with the Intergovernmental Panel on Climate Change Assessment Report 6 guidelines (Masson-Delmotte et al., 2021). It has to be noted that the baseline period is not a stationary period since it includes for example, the strong trend observed in the second half of the 20th century due to the significant influence of aerosols. For further insight in the timeseries of mean ISM rainfall refer to Katzenberger et al. (2021). The 50-years period is chosen in order to obtain statistically robust results. The results for the stationary preindustrial baseline period 1900-1950 as well as the shorter period 1985-2015 provide comparable results (See Tables S2 in Supporting Information S1).

2.2. Definitions

Observations have shown that even in seasons that exceed the longterm rainfall averages by 10% (which corresponds to one standard deviation of longterm seasonal precipitation; Singh et al. (2019)), severe floods on large scales with significant impact on the Indian food production and the Gross Domestic Product occur (Gadgil & Gadgil, 2006; Wang et al., 2015). Thus, some studies define surplus/excess/flood/wet years as the longterm average ISM rainfall plus 10% of the mean (Mohanty et al., 2002; Parthasarathy et al., 1994; Singh et al., 2019). To focus on the even more extreme monsoon seasons, we define very wet monsoons seasons by the 90%-percentile in this study. This means that we classify a monsoon season as very wet if the seasonal rainfall falls within the range of rainfall during the five wettest seasons in 1965–2015, thus what corresponds to a 1-in-10-years monsoon season. These very wet seasons can be considered as years with a very high risk for severe floods and socioeconomic impacts. For the analysis on the subseasonal scale, we classify the daily rainfall amount following the definitions applied by Sharmila et al. (2015): Wet days are defined as days with rainfall of at least 0.1 mm/day and light rainfall as precipitation between 0.1 and 10 mm/day. Moderate rainfall ranges from 10 to 40 mm/day and heavy rainfall is defined as precipitation exceeding 40 mm/day. We also calculated the Simple Daily Intensity Index (SDII) in order to get an insight on the mean precipitation amount at rainy days (>1 mm).

2.3. Model Comparison

Rajendran et al. (2021) provide a detailed study about the evaluation of the CMIP6 models in the context of the ISM. The selection criteria are based on the seasonal monsoon rainfall, the interannual variability but also aspects like the spatial distribution. Applying these criteria for 1951–2010 and using gridded rainfall data of the Indian Meteorological Department (Pai et al., 2014), six of the 32 models available for this study are classified to have the best historic monsoon performance in their study. These models are considered with particular focus in our study.

3. Results

In order to understand the changes in the seasonal precipitation between 1965 and 2015 and the second half of the 21st century, we analyze the former and future distribution of yearly JJAS mean precipitation (Figure 1). Under the SSP5-8.5 scenario, the multi-model average shows an increase in the seasonal mean precipitation (6.8 mm/day) as compared to the historical period (5.6 mm/day). Taking only the models with a reasonably good representation of the ISM into consideration, the increase is from 6.2 mm/day to 7.3 mm/day. In the modest mitigation scenario (SSP2-4.5), the historic seasonal mean increases to 6.3 mm/day averaging over all models and



Geophysical Research Letters





in SSP1-2.6 the increase is to 6.2 mm/day. This change is accompanied by an increase in variation of seasonal monsoon precipitation: For all models, the standard deviation increases in SSP5-8.5, which is associated with an increase of the year-to-year (interannual) variability.

By definition of the 90%-percentile with regard to the baseline period 1965–2015, five extraordinary wet seasons have exceeded the threshold in this past period. For the future period 2050–2100, the general shift of the precipitation distribution toward wetter seasons is accompanied by more seasons exceeding this threshold. Accordingly, dry seasons are projected to be rarer in the 21st century (only taking into account rainfall; for water availability the role of evaporation is also crucial which was beyond the focus of this study). The temporal development toward more severely wet seasons is displayed in Figure 2a. The variation is within multi-model mean plus one standard deviation (based on the historical CMIP6 data ranging from 1950 to 2015) for 1950–2027, but a significant increase beyond the observed natural variability emerges in 2027, which accelerates non-linearly. The spatial distribution of the 90th-percentile of the mean seasonal rainfall in 1965–2015 is displayed in Figure 2b. The regions in which the monsoon seasonal rainfall is projected to intensify most is shown in Figure 2c. Particularly the Himalayan region, Bangladesh, the west coast and central India are projected to intensify most. The number of seasons in the period of 2050–2100 that are classified as very wet according to our definition, can be seen at





Figure 2. (a.) Increase of very wet Indian summer monsoon (ISM) seasons (JJAS) in the 21st century under unabated climate change (SSP5-8.5). For each of the 32 models in our study, the number of very wet seasons in the preceding 50-year period is calculated (1965–2015, 1966–2016, ..., 2051–2100), and the resulting timeseries are smoothed with a 10 years moving average. For color explanation refer to Figure 1. The red curve marks the median of all models. The blue curve marks the median of the models with better ISM representation. The horizontal gray line marks the reference from 1965 to 2015 when per definition 5 out of 50 seasons were very wet. 90th-percentile based on mean seasonal rainfall in 1965–2015 (b.) and projected change in the 90th-percentile between 1965–2015 and 2050–2100 under SSP5-8.5 (c.) for the multi-model mean of six models with best monsoon performance.

the very right of the Figure 2a and in the circular bar plot in Figure 3. The multi-model mean of very wet seasons in 2050–2100 is 38.9 under SSP5-8.5 (thus 7.8 times more than in 1965–2015), 29.1 under SSP2-4.5 (factor 5.8) and 25.1 under SSP1-2.6 (factor 5.0). Results for individual models are given in Table S1 of Supporting Information S1. The models with a better representation of the ISM in the historical simulations project 39.5 very wet seasons in 2050–2100 under SSP5-8.5, 24.1 seasons under SSP2-4.5 and 22.1 seasons under SSP1-2.6.

The relationship between the increase in very wet seasons under SSP5-8.5 and the increase in global mean temperature is - apart from some few outliers - relatively linear (see Figure S2 in Supporting Information S1; with outliers: correlation coefficient = 0.12). The slope is 9.6 seasons/K, which means that per degree of global warming, it is projected that within 50-year-periods, 9.6 more summer monsoon seasons are to be very wet.

The results for the changes in the amount of daily precipitation are displayed in Figure 4. The spatial distribution for the individual models can be found in the Figures S3 and S4 of Supporting Information S1. The number of wet days (>0.1 mm) is projected to increase in most of the Indian monsoon region. Interestingly, most of the general circulation models coincide in an increase of wet days particularly in the Northwest of India. Nevertheless, there still remains a considerable range between the intensity of the change in this area. The number of days with light rainfall (10 mm > x > 0.1 mm) in each season is projected to decrease in multi-model average, particularly in Bangladesh and North of the Himalayan mountains as well as the West coast of India. In Pakistan, the number is projected to increase. Regarding moderate rainfall days (40 mm > x > 10 mm), the multi-model mean suggests an increase in most areas, while in Bhutan and South Pakistan a decrease is projected. However, the individual models show a broad range of different patterns with limited overlap between the models. For heavy rainfall days (>40 mm), an increase in the order of 0–2.5 days per season is projected, specially in the Westen Ghats, the South coast of Bangladesh and the Himalayan region. These results coincide with the results for the SDII that show an intensification of the mean precipitation amount on rainy days (Figure S5 in Supporting Information S1).

4. Discussion

In this study, we use 32 CMIP6 models to analyze the future frequency of very strong rainfall seasons under climate change. We found a clear, general increase of the yearly ISM rainfall between 1965-2015 and 2050–2100 toward wetter seasons. Under SSP5-8.5, the increasing trend becomes significant in 2027. Similar intensifying





Figure 3. Increase in the number of very wet Indian summer monsoon (ISM) seasons by the second half of the 21st century in comparison to 1965–2015 in the Indian monsoon region under sustainable development (SSP1-2.6), modest mitigation (SSP2-4.5) and unabated climate change (SSP5-8.5). Each circle segment represents the result for one model. A full segment means that 50 out of 50 seasons in the future period of 2050–2100 are projected to be very wet (compared to the 1965–2015 baseline period). Blue bars represent models with a better representation of the ISM in the historical simulations, while orange bars belong to the models with less ability of simulating the ISM in historical simulations. The cyan circle represents the reference period of 1965–2015, in which by definition 5 seasons exceed the 90%-percentile of that period. For GFDL-CM4 (GC) in SSP1-2.6 no data was available. For model abbreviations refer to Figure 1.





Figure 4. Change in (a.) wet days (>0.1 mm) and days with (b.) light (10 mm > x > 0.1 mm), (c.) moderate (10 mm > x > 40 mm) and (d.) heavy (>40 mm) rainfall between 1965–2015 and 2050–2100 under SSP5-8.5. The change is shown as number of days per season that change in average between the two periods. The multi-model mean of the six models with best monsoon performance in the historical simulations is displayed. In (e.) the results for the single models as well as the multi-model mean are summarized (mean from monsoon region). More details regarding the spatial distribution of the change are displayed in the Figures S3 and S4 of Supporting Information S1.

tendencies of very wet seasons were found by other studies, but partly were not yet able to provide statistically robust results (Kamizawa & Takahashi, 2018) or still had a large spread between the models selected (Sharmila et al., 2015; Turner & Annamalai, 2012). Sharmila et al. (2015) used the 4 CMIP5 models that captured monsoon rainfall features best and found that strong monsoons might get wetter under the strongest emission scenario. This might be due to a projected increase in persistent wet spells and a decrease in dry spells in strong monsoon years (Sharmila et al., 2015). In the literature, these terms are commonly referred to as active and break spells (Singh et al., 2014). The spells of the Indian monsoon have their origin in the interaction between multiple modes of propagating intraseasoal oscillations consisting of 10–20 days and 30–50 days fluctuations (Krishnamurthy & Shukla, 2007). In observations, an increase of the intensity of wet spells between 1951-1980 and 1981–2011 was found by a previous study (Singh et al., 2014). However, changes in frequency, duration and extent are still under scientific debate (Singh et al., 2019).

The increasing tendency of very wet seasons is linked to the projected increase of the mean ISM rainfall amount examined and confirmed by several studies (Chaturvedi et al., 2012; Katzenberger et al., 2021; Lee & Wang, 2014; Mei et al., 2015; Menon et al., 2013; Sharmila et al., 2015). There is a widespread agreement (Singh et al., 2019) that this trend is the consequence of the Indian Ocean's rising temperature increasing atmospheric moisture content and thus moisture convergence (Cherchi et al., 2011; Mei et al., 2015; Seth et al., 2013; Sooraj et al., 2015). In addition to this increased amount of precipitation in June-September, an increase in interannual variability of seasonal rainfall is projected (Jayasankar et al., 2015; Katzenberger et al., 2021; Kitoh, 2017; Menon et al., 2013). Since the occurrence of El-Niño and La-Niña years are linked to weak and strong ISM years (Kumar et al., 2006), the El Niño Southern Oscillation (ENSO) might contribute to this trend in interannual variability. Nevertheless, no consensus regarding the future of ENSO activity, for example, regarding frequency and amplitude has been reached (Cai et al., 2015; Chen et al., 2017; Collins et al., 2010).

Given that Dong and Dong (2021) found an improvement of the representation of daily rainfall with more than 10 mm and more than 20 mm, the results of CMIP6 regarding the daily amount of rainfall are of particular interest. In multi-model mean, we found a decrease of light rainfall and an increase of heavy rainfall days in a large area of the Indian monsoon region confirming the results of a previous study based on CMIP5 models (Sharmila et al., 2015). While the results of Sharmila et al. (2015) indicate a rather decrease of days with moderate rainfall, the CMIP6 models in our study tend to project an increase in moderate rainfall days. Simultaneously, the spatial pattern of the projected changes is quite similar in CMIP6 and CMIP5. However, particularly regarding the light rainfall days, there also remains a disparity between models in CMIP6. These differences are also a result of different wet or dry bias in the individual models, in both Sharmila et al. (2015) as well as this study.

Regarding the number of wet days during a monsoon season, we found an increasing tendency toward future monsoon seasons. This result is opposing the result of Sharmila et al. (2015) who reported a decrease in the multi-model mean. But as the authors of this study also pointed out, this result is based on the spatial pattern of four CMIP5 models with a rather limited overlap between the tendencies in the individual models. In comparison, the results from the CMIP6 models show less spread between the individual models and coincide also regarding the strong increasing tendency in the Northwest of the monsoon region. The difference between CMIP6 and CMIP5 might be due to the different climate models underlying the analysis, but could also be a result of the improvements between the model generations.

By using only one ensemble member per model in this study, we neglected the possible contribution of internal model variability to the uncertainty of the model's projection. Besides, as some model centers provide several models and certain model components are used among different modeling groups, the models cannot be regarded independent from each other (Knutti et al., 2017).

5. Conclusion

We used 32 of the latest generation of climate models in CMIP6 to update the findings on very wet seasons from CMIP5. For this purpose, we analyze the change in very wet ISM seasons between 1965–2015 and 2050–2100 under unabated (SSP5-8.5), modestly abated (SSP2-4.5) as well as strongly abated (SSP1-2.6) climate change. Under all scenarios, we found a clear change toward wetter seasons between the two periods which is associated with an increase in severity as well as frequency of very strong rainfall seasons. This increase is less clear under SSP2-4.5 and SSP1-2.6. Under the same scenario, very wet seasons that used to occur in 5 out of 50 years in 1965–2015 are projected on average to appear in 39 out of 50 years in 2050–2100. Under SSP2-4.5, they are projected to occur in 29 out of 50 years. Thus, even modest efforts to mitigate climate change can have a strong impact on the frequency of very strong rainfall seasons.

In this study, we also set the focus on the question how this intensification of very wet ISM seasons translates to the subseasonal scale, in particular the amount of daily rainfall. In this context, we found that the CMIP6 models project a shift from days with light rainfall days toward days with moderate or heavy rainfall. This intensification of daily rainfall can also be seen in the increase of the SDII. However, the CMIP6 models remain with a considerable multi-model spread for some regions. Regarding the number of wet days, the models largely coincide in projecting an increasing tendency, particularly for the Northwest of the monsoon region.

While an increase (decrease) in summer monsoon rainfall is associated with an increase (decrease) in crop yields, this relationship is not valid for precipitation rainfalls exceeding or falling below a certain threshold (Kumar et al., 2004; Prasanna, 2014). Varghese et al. (2020) projected a decrease in future rice yields in Kerala as a result of a rise in temperature and a decrease in rainfall in the region that in combination reduces the maturity time of the crop as well as the total biomass production. Revadekar and Preethi (2012) found that fluctuations in various indices of precipitation extremes of the ISM are related to foodgrain yields. Considering additionally the fact that the population of India is expected to further grow and to reach 1.5 billion in the mid 21st century (Dyson et al., 2005), the increased frequency of very strong ISM rainfall seasons found in this study has to remain object of further studies in order to implement adaptions strategies in agriculture and not to become a serious threat to food security in India.

Data Availability Statement

The datasets from CMIP6 simulations are available via the CMIP6 Search Interface: https://esgf-node.llnl.gov/search/cmip6/ (last access: 31 March 2022) (WCRP, 2022). The codes for this study are permanently available: https://doi.org/10.5281/zenodo.6973330 (Katzenberger, 2022).

Acknowledgments

The authors acknowledge the World Climate Research Programme's Working Group on Coupled Modelling, which is responsible for CMIP, and we thank the climate modeling groups for producing and making available their model output. The research was funded by the Heinrich-Boell Foundation. The publication of this article was funded by the Open Access Fund of the Leibniz Association. Open access funding enabled and organized by Projekt DEAL.

References

- Almazroui, M., Saeed, F., Saeed, S., Ismail, M., Ehsan, M. A., Islam, M. N., et al. (2021). Projected changes in climate extremes using CMIP6 simulations over SREX regions. *Earth Systems and Environment*, 5(3), 481–497. https://doi.org/10.1007/s41748-021-00250-5
- Almazroui, M., Saeed, S., Saeed, F., Islam, M. N., & Ismail, M. (2020). Projections of precipitation and temperature over the South Asian countries in CMIP6. *Earth Systems and Environment*, 4(2), 297–320. https://doi.org/10.1007/s41748-020-00157-7
- Bohra, A., Basu, S., Rajagopal, E., Iyengar, G., Gupta, M. D., Ashrit, R., & Athiyaman, B. (2006). Heavy rainfall episode over Mumbai on 26 July 2005: Assessment of NWP guidance. *Current Science*, 90(9), 1188–1194. https://www.jstor.org/stable/24092019
- Cai, W., Santoso, A., Wang, G., Yeh, S.-W., An, S.-I., Cobb, K. M., et al. (2015). ENSO and greenhouse warming. *Nature Climate Change*, 5(9), 849–859. https://doi.org/10.1038/nclimate2743
- Chaturvedi, R. K., Joshi, J., Jayaraman, M., Bala, G., & Ravindranath, N. (2012). Multi-model climate change projections for India under representative concentration pathways. *Current Science*, 103(7), 791–802. http://www.jstor.org/stable/24088836
- Chen, L., Li, T., Yu, Y., & Behera, S. K. (2017). A possible explanation for the divergent projection of ENSO amplitude change under global warming. *Climate Dynamics*, 49(11), 3799–3811. https://doi.org/10.1007/s00382-017-3544-x
- Cherchi, A., Alessandri, A., Masina, S., & Navarra, A. (2011). Effects of increased CO₂ levels on monsoons. *Climate Dynamics*, 37(1–2), 83–101. https://doi.org/10.1007/s00382-010-0801-7
- Collins, M., An, S.-I., Cai, W., Ganachaud, A., Guilyardi, E., Jin, F.-F., et al. (2010). The impact of global warming on the tropical Pacific Ocean and El Niño. *Nature Geoscience*, *3*(6), 391–397. https://doi.org/10.1038/ngeo868
- DeFries, R., Mondal, P., Singh, D., Agrawal, I., Fanzo, J., Remans, R., & Wood, S. (2016). Synergies and trade-offs for sustainable agriculture: Nutritional yields and climate-resilience for cereal crops in Central India. *Global Food Security*, 11, 44–53. https://doi.org/10.1016/j. gfs.2016.07.001
- Dong, T., & Dong, W. (2021). Evaluation of extreme precipitation over Asia in CMIP6 models. *Climate Dynamics*, 57(7), 1751–1769. https:// doi.org/10.1007/s00382-021-05773-1
- Dyson, T., Cassen, R., & Visaria, L. (2005). Twenty-first century India: Population, economy, human development, and the environment. OUP Catalogue, 32(2), 369–372. https://doi.org/10.1002/psp.418
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the coupled model intercomparison project phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9(5), 1937–1958. https://doi. org/10.5194/gmd-9-1937-2016
- Gadgil, S., & Gadgil, S. (2006). The Indian monsoon, GDP and agriculture. Economic and Political Weekly, 4887–4895. http://www.jstor.org/ stable/4418949
- Guha-Sapir, D., Below, R., & Hoyois, P. (2018). EM-DAT: The CRED/OFDA international disaster database. Université Catholique de Louvain. Retrieved from https://www.emdat.be
- Gusain, A., Ghosh, S., & Karmakar, S. (2020). Added value of CMIP6 over CMIP5 models in simulating Indian summer monsoon rainfall. Atmospheric Research, 232, 104680. https://doi.org/10.1016/j.atmosres.2019.104680
- Ha, K.-J., Moon, S., Timmermann, A., & Kim, D. (2020). Future changes of summer monsoon characteristics and evaporative demand over Asia in CMIP6 simulations. *Geophysical Research Letters*, 47(8), e2020GL087492. https://doi.org/10.1029/2020gl087492
- Hunt, K. M., & Menon, A. (2020). The 2018 Kerala floods: A climate change perspective. *Climate Dynamics*, 54(3), 2433–2446. https://doi.org/10.1007/s00382-020-05123-7
- Jayasankar, C., Surendran, S., & Rajendran, K. (2015). Robust signals of future projections of Indian summer monsoon rainfall by IPCC AR5 climate models: Role of seasonal cycle and interannual variability. *Geophysical Research Letters*, 42(9), 3513–3520. https://doi. org/10.1002/2015GL063659
- Kamizawa, N., & Takahashi, H. G. (2018). Projected trends in interannual variation in summer seasonal precipitation and its extremes over the tropical Asian monsoon regions in CMIP5. Journal of Climate, 31(20), 8421–8439. https://doi.org/10.1175/jcli-d-17-0685.1
- Katzenberger, A. (2022). Very wet monsoon seasons in India under global warming available code [zenodo]. https://doi.org/10.5281/ zenodo.6973330
- Katzenberger, A., Schewe, J., Pongratz, J., & Levermann, A. (2021). Robust increase of Indian monsoon rainfall and its variability under future warming in CMIP6 models. *Earth System Dynamics*, 12(2), 367–386. https://doi.org/10.5194/esd-12-367-2021
- Kitoh, A. (2017). The Asian monsoon and its future change in climate models: A review. Journal of the Meteorological Society of Japan. Ser. II, 95(1), 7–33. https://doi.org/10.2151/jmsj.2017-002
- Knutti, R., Sedláček, J., Sanderson, B. M., Lorenz, R., Fischer, E. M., & Eyring, V. (2017). A climate model projection weighting scheme accounting for performance and interdependence. *Geophysical Research Letters*, 44(4), 1909–1918. https://doi.org/10.1002/2016GL072012
- Krishnamurthy, V., & Shukla, J. (2007). Intraseasonal and seasonally persisting patterns of Indian monsoon rainfall. *Journal of Climate*, 20(1), 3–20. https://doi.org/10.1175/JCLI3981.1
- Kumar, K. K., Kumar, R. K., Ashrit, R., Deshpande, N., & Hansen, J. W. (2004). Climate impacts on Indian agriculture. International Journal of Climatology: A Journal of the Royal Meteorological Society, 24(11), 1375–1393. https://doi.org/10.1002/joc.1081
- Kumar, K. K., Rajagopalan, B., Hoerling, M., Bates, G., & Cane, M. (2006). Unraveling the mystery of Indian monsoon failure during El Niño. Science, 314(5796), 115–119. https://doi.org/10.1126/science.1131152
- Lee, J.-Y., & Wang, B. (2014). Future change of global monsoon in the CMIP5. Climate Dynamics, 42(1–2), 101–119. https://doi.org/10.1007/ s00382-012-1564-0
- Masson-Delmotte, P. V., Zhai, A., Pirani, S. L., Connors, C., Péan, S., Berger, N., et al. (2021). IPCC: Climate change 2021: The physical science basis. Contribution of working group I to the sixth assessment report of the intergovernmental panel on climate change. Cambridge University Press. https://www.ipcc.ch/report/ar6/wg1/
- Mei, R., Ashfaq, M., Rastogi, D., Leung, L. R., & Dominguez, F. (2015). Dominating controls for wetter South Asian summer monsoon in the twentyfirst century. *Journal of Climate*, 28(8), 3400–3419. https://doi.org/10.1175/jcli-d-14-00355.1
- Menon, A., Levermann, A., Schewe, J., Lehmann, J., & Frieler, K. (2013). Consistent increase in Indian monsoon rainfall and its variability across CMIP-5 models. *Earth System Dynamics*, 4(2), 287–300. https://doi.org/10.5194/esd-4-287-2013
- Mohanty, U., Bhatla, R., Raju, P., Madan, O., & Sarkar, A. (2002). Meteorological fields variability over the Indian seas in pre and summer monsoon months during extreme monsoon seasons. *Journal of Earth System Science*, 111(3), 365–378. https://doi.org/10.1007/bf02701981
- Mujumdar, M., Preethi, B., Sabin, T., Ashok, K., Saeed, S., Pai, D., & Krishnan, R. (2012). The Asian summer monsoon response to the La Nina event of 2010. *Meteorological Applications*, 19(2), 216–225. https://doi.org/10.1002/met.1301
- O'Neill, B. C., Tebaldi, C., Van Vuuren, D. P., Eyring, V., Friedlingstein, P., Hurtt, G., et al. (2016). The scenario model intercomparison project (ScenarioMIP) for CMIP6. *Geoscientific Model Development*, 9, 3461–3482. https://doi.org/10.5194/gmd-9-3461-2016



- Pai, D., Rajeevan, M., Sreejith, O., Mukhopadhyay, B., & Satbha, N. (2014). Development of a new high spatial resolution (0.25×0.25) long period (1901–2010) daily gridded rainfall data set over India and its comparison with existing data sets over the region. *Mausam*, 65(1), 1–18. https://doi.org/10.54302/mausam.v65i1.851
- Parthasarathy, B., Munot, A., & Kothawale, D. (1994). All-India monthly and seasonal rainfall series: 1871–1993. Theoretical and Applied Climatology, 49(4), 217–224. https://doi.org/10.1007/bf00867461
- Prasanna, V. (2014). Impact of monsoon rainfall on the total foodgrain yield over India. Journal of Earth System Science, 123(5), 1129–1145. https://doi.org/10.1007/s12040-014-0444-x
- Rajendran, K., Surendran, S., Varghese, S. J., & Sathyanath, A. (2021). Simulation of Indian summer monsoon rainfall, interannual variability and teleconnections: Evaluation of CMIP6 models. *Climate Dynamics*, 58(9–10), 1–31. https://doi.org/10.1007/s00382-021-06027-w
- Revadekar, J., & Preethi, B. (2012). Statistical analysis of the relationship between summer monsoon precipitation extremes and foodgrain yield over India. *International Journal of Climatology*, 32(3), 419–429. https://doi.org/10.1002/joc.2282
- Seth, A., Rauscher, S. A., Biasutti, M., Giannini, A., Camargo, S. J., & Rojas, M. (2013). CMIP5 projected changes in the annual cycle of precipitation in monsoon regions. *Journal of Climate*, 26(19), 7328–7351. https://doi.org/10.1175/JCLI-D-12-00726.1
- Sharmila, S., Joseph, S., Sahai, A. K., Abhilash, S., & Chattopadhyay, R. (2015). Future projection of Indian summer monsoon variability under climate change scenario: An assessment from CMIP5 climate models. *Global and Planetary Change*, 124, 62–78. https://doi.org/10.1016/j. gloplacha.2014.11.004
- Singh, D., Ghosh, S., Roxy, M. K., & McDermid, S. (2019). Indian summer monsoon: Extreme events, historical changes, and role of anthropogenic forcings. Wiley Interdisciplinary Reviews: Climate Change, 10(2), 1–35. https://doi.org/10.1002/wcc.571
- Singh, D., Tsiang, M., Rajaratnam, B., & Diffenbaugh, N. S. (2014). Observed changes in extreme wet and dry spells during the South Asian summer monsoon season. *Nature Climate Change*, 4(6), 456–461. https://doi.org/10.1007/s00382-019-04997-6
- Sooraj, K., Terray, P., & Mujumdar, M. (2015). Global warming and the weakening of the Asian summer monsoon circulation: Assessments from the CMIP5 models. *Climate Dynamics*, 45(1–2), 233–252. https://doi.org/10.1007/s00382-014-2257-7
- Tayler, K., Stouffer, R., & Meehl, G. (2012). An overview of CMIP5 and the experimental design. *Bulletin of the American Meteorological Society*, 93, 485–498. https://doi.org/10.1175/BAMS-D-11-00094.1
- Turner, A. G., & Annamalai, H. (2012). Climate change and the South Asian summer monsoon. *Nature Climate Change*, 2(8), 587–595. https://doi.org/10.1038/nclimate1495
- Van Vuuren, D. P., Kriegler, E., O'Neill, B. C., Ebi, K. L., Riahi, K., Carter, T. R., et al. (2014). A new scenario framework for climate change research: Scenario matrix architecture. *Climatic Change*, 122(3), 373–386. https://doi.org/10.1007/s10584-013-0906-1
- Varghese, J., Surendran, S., Ajithkumar, B., Rajendran, K., & Kitoh, A. (2020). Future changes in rice yield over Kerala using climate change scenario from high resolution global climate model projection. *Journal of Earth System Science*, 129(1), 192. https://doi.org/10.1007/ s12040-020-01459-0
- Vishnu, C., Sajinkumar, K., Oommen, T., Coffman, R., Thrivikramji, K., Rani, V., & Keerthy, S. (2019). Satellite-based assessment of the August 2018 flood in parts of Kerala, India. *Geomatics, Natural Hazards and Risk, 10*(1), 758–767. https://doi.org/10.1080/19475705.2018.1543212
- Wang, B., Xiang, B., Li, J., Webster, P. J., Rajeevan, M. N., Liu, J., & Ha, K.-J. (2015). Rethinking Indian monsoon rainfall prediction in the context of recent global warming. *Nature Communications*, 6(1), 1–9. https://doi.org/10.1038/ncomms8154

WCRP. (2022). CMIP6 data. Retrieved from https://esgf-node.llnl.gov/search/cmip6/